Week 28 starting 17 July 06
SM18 Operation Meeting (Tuesdays) 18 July 06

Bench status  on Monday  17-07-2006 at:   7/17/2006 9:27:17 AM  
	9
	A1
	MBAL1396
	PT 13.2 Resist. Meas.
	301.35

	12
	A2
	B. Repair
	[image: image1.png]



	300.28

	8
	B1
	SSS142
	ICS 3 Traveller after warm test
	301.89

	2
	B2
	SSS275
	PT 12 Warm Up
	1.90

	7
	C1
	SSS138
	PT 3 QH Measure
	2.35

	3
	C2
	SSS266
	PT 6.6 Training 6
	1.91

	10
	D1
	SSS636
	MTM Mech non-avail
	300.01

	1
	D2
	SSS645
	PT 6.6 Training 6
	4.51 HV

	6
	E1
	MBAR1326
	ICS 4 Final connection
	302.6

	5
	E2
	MBAL2353
	PT 13.1 IAP @ Warm
	298.04

	11
	F1
	MBBR2296
	PT 13.2 Resist. Meas.
	299.65

	4
	F2
	MBAL2349
	PT 8 Fast De-excitation
	1.92


  CONFIG :  D 6 - AS 3 -  IRSSS 3
Supervisor: Gijs (Monday) ; Pierre after;Eq support : 16-3602 
15 Magnet Tests completed : 
8 Dipoles (2290  with Magnetic Mesurment), 
5 Arc-SSS  (271 3 times thermal cycle)

0  500-series [ ]
2  600-series [ 676 (166 hrs), 682 after TH Cy (150 hrs) ]
LOT of StandBy’s & lot of HV problems (Real + humidity etc) this week

682: 6 MQTL’s ; probable open circuit in connection between 4 & 5
General Points :
· Situation from Cryogenics:  J-P Lamboy or ACR Rep.
· News /Situation from (SSS, SSSS)  :  J Halik
· Situation SSSS  :OP issues : 
· A.O.B..
	Magnets under Test from 09-07-2006 to 15-07-2006 

	-
	Bench
	Magnet
	Shafts
	+
	Start Date
	Connect
Time
	Cooling
Time
	Cold Test
Time
	Warming
Time
	Disconnect
Time
	End Date
	Total
	Signed for
	
	

	1
	TBA1
	MBBR1306 
	[image: image2.png]



	


	06-07-2006 
	41 h.
	37 h.
	64 h.
	15 h.
	13 h.
	13-07-2006 
	170 h.
	Stripping
	
	

	2
	TBA2
	MBBR2290 
	[image: image4.png]



	


	06-07-2006 
	29 h.
	29 h.
	16 h.
	17 h.
	26 h.
	11-07-2006 
	117 h.
	Stand By::HV CQBP
	
	

	3
	TBB1
	SSS258 
	[image: image6.png]



	


	07-07-2006 
	30 h.
	14 h.
	16 h.
	8 h.
	26 h.
	11-07-2006 
	94 h.
	Stripping
	
	

	4
	TBB1
	SSS262 
	[image: image8.png]



	


	11-07-2006 
	33 h.
	16 h.
	23 h.
	9 h.
	20 h.
	15-07-2006 
	101 h.
	Stripping
	
	

	5
	TBB2
	SSS676 
	[image: image10.png]



	


	05-07-2006 
	69 h.
	24 h.
	27 h.
	11 h.
	35 h.
	12-07-2006 
	166 h.
	Stand By
	
	

	6
	TBC1
	SSS047 
	[image: image12.png]



	


	04-07-2006 
	22 h.
	14 h.
	72 h.
	6 h.
	8 h.
	09-07-2006 
	122 h.
	Stand By: HV CQAP
	
	

	7
	TBC1
	SSS274 
	[image: image14.png]



	


	09-07-2006 
	36 h.
	24 h.
	30 h.
	6 h.
	13 h.
	14-07-2006 
	109 h.
	Stand By :TT821
	
	

	8
	TBC2
	SSS141 
	[image: image16.png]



	


	08-07-2006 
	33 h.
	16 h.
	46 h.
	7 h.
	13 h.
	14-07-2006 
	115 h.
	Stand By Corr Pwrng
	
	

	9
	TBD2
	SSS682 
	[image: image18.png]



	


	06-07-2006 
	2 h.
	22 h.
	86 h.
	14 h.
	26 h.
	12-07-2006 
	150 h.
	Stand By: MQTL repair
	
	

	10
	TBE1
	MBAL2354 
	[image: image20.png]



	


	10-07-2006 
	45 h.
	24 h.
	32 h.
	15 h.
	12 h.
	15-07-2006 
	128 h.
	Stripping
	
	

	11
	TBE1
	MBBR1315 
	[image: image22.png]



	


	06-07-2006 
	27 h.
	22 h.
	15 h.
	15 h.
	11 h.
	10-07-2006 
	90 h.
	Stand By  :TT821
	
	

	12
	TBE2
	MBBR1349 
	[image: image24.png]



	


	08-07-2006 
	24 h.
	19 h.
	23 h.
	15 h.
	10 h.
	12-07-2006 
	91 h.
	Stripping
	
	

	13
	TBF1
	MBAL1397 
	[image: image26.png]



	


	06-07-2006 
	58 h.
	26 h.
	17 h.
	16 h.
	25 h.
	12-07-2006 
	142 h.
	Stripping
	
	

	14
	TBF2
	MBAL1398 
	[image: image28.png]



	


	09-07-2006 
	28 h.
	25 h.
	19 h.
	23 h.
	18 h.
	14-07-2006 
	113 h.
	Stripping
	
	

	15
	TBF2
	MBBR1356 
	[image: image30.png]



	


	05-07-2006 
	25 h.
	25 h.
	20 h.
	14 h.
	14 h.
	09-07-2006 
	98 h.
	Stripping
	
	


	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBBR1306
	11036.2 A.
(D1_U)
	11370.5 A.
(D1_L)
	11918.4 A.
(D2-L)
	12503 A.
(D1-U)
	12779.9 A.
(D2_L)
	12850 A.
[image: image32.jpg]



	-
	-
	-
	-
	-
	-
	-

	2
	MBBR2290
	12216.5 A.
(D1_U)
	12771.2 A.
(D2_U)
	12852 A.
[image: image33.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	3
	SSS258
	12105.6 A.
(QE_12)
	12552 A.
(QE_12)
	12728 A.
(QI-34)
	[image: image34.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	4
	SSS262
	11188.9 A.
(QE_34)
	12032.8 A.
(QE_34)
	12000 A.
(No Data)
	12370 A.
(QE-34)
	[image: image35.png]



	-
	-
	-
	-
	-
	-
	-
	-

	5
	SSS676
	5491 A.
(Ext-34)
	5820 A.
[image: image36.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	SSS047
	10863 A.
(QI_34)
	11131.2 A.
(QE_12)
	11509.7 A.
(QE_34)
	11893.2 A.
(QI_12)
	12501.5 A.
(QE_12)
	12850 A.
[image: image37.jpg]



	-
	-
	-
	-
	-
	-
	-

	7
	SSS274
	12110.9 A.
(QI_12)
	12421 A.
(QE_12)
	12676.9 A.
(QI_12)
	[image: image38.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	8
	SSS141
	11579.1 A.
(QI_12)
	11914.8 A.
(QE_12)
	12163.7 A.
(QI_12)
	12337.2 A.
(QI-12)
	12756 A.
(QI-12)
	12850.4 A.
(Q1_12)
	12850 A.
(Vclamp E)
	12850 A.
(Vbus_Ext)
	12850 A.
[image: image39.jpg]



	-
	-
	-
	-

	9
	SSS682
	[image: image40.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	10
	MBAL2354
	10281.6 A.
(D1_L)
	11409.3 A.
(D2_L)
	11880.7 A.
(D1_L)
	12450 A.
(D2-U)
	12850 A.
[image: image41.jpg]



	-
	-
	-
	-
	-
	-
	-
	-

	11
	MBBR1315
	12613.4 A.
(D2_U)
	12850 A.
[image: image42.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	12
	MBBR1349
	11323.2 A.
(D1_U)
	12301.5 A.
(D1_L)
	12294.1 A.
(D1-L)
	0 A.
(No Data)
	[image: image43.png]



	-
	-
	-
	-
	-
	-
	-
	-

	13
	MBAL1397
	12446.6 A.
(D2- L)
	12612.1 A.
(D1 - L)
	0 A.
(No Data)
	12805.9 A.
(D1_L)
	[image: image44.png]



	-
	-
	-
	-
	-
	-
	-
	-

	14
	MBAL1398
	12359.2 A.
(D2_L)
	12317.5 A.
(D2_U)
	12517.1 A.
(D2_U)
	[image: image45.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	15
	MBBR1356
	11182 A.
(D1_L)
	12212.8 A.
(D2-U)
	12146.3 A.
(D2-L)
	[image: image46.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-


[image: image47.png]Magnets from Sunday, July 09, 2006 to Saturday, July 15, 2006

- .-.- [ -..l

170

Ref 18 136 187 13 234 135 862 SI4l S04 924 Sere S8 52 290 1306





1306 took long because of cohabitation with 2290 : both on A cluster at same time;682 : 6 MQTL’s  2nd run had problem ; 1397 long connect time ? 

STATISTICS for endJune 2006
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	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Tuesday_11_M

Tuesday_11_A
	TBD1
	SSS636 : MM1+MM2+MM3

-Problem with_T_coil : no power supply. 

-Power supply of T_coil need to be configured

-T_coil: One temperature out of range.

- MM1 is not correct

-MM2, MM3 results are displaying sectors in over-ranges.


	MTM
	
	Power supply taken from another bench.

Richard advised to proceed.

Check by Stephane and Richard. Not correct config_files. For MQY single, test “family_3” config files have to be selected.

	Tuesday_11_N
	TBA2
	MB2347: file of PREP2.1 (IAP@warm LF heater discharge) saved in 2347_2 folder due to mistaken selection in TEMA
	MTM
	
	 Eq sup contacted Richard copied the file in the correct folder.

	Tuesday_11_N
	TBB1
	SSS262: DAQ (LF) did not trigger in Prep2.2C (clamp connection check). 
	MTM
	
	Richard contacted. He informed POTAIM card was adjusted for previous magnet (without CD and EF circuits) hence LF to be triggered manually. Test conducted with manual trigger

	Tuesday_11_N
	TBD1
	SSS636: Over range errors in sector S0405 of Ap2 in MM2.
	MTM
	
	Richard in consultation with Stephane suggested to validate this in MM2 and MM3 as the concerned sector is in middle and extrapolation is feasible. 

	Tuesday_11_N
	TBF2
	MB1398: PT4 (ramp to 1000A) could not be performed. Power application shows an error '222 at serial port checking.' and interlock 'VS_FAULT' remains after reset. Attempt was made with reset of server PLC as well as Power converter without success.

	MTM
	
	Richard contacted.

VS_FAULT is cleared after switching off excess water connections (limit of 3 power supplies at the same time).

	Tuesday_11_N
	TBC2
	SSS141: Quench occurred two times at ultimate (in the clamp). 2nd quench occurred after holding 12850 for 5min and shown by AQA outside the magnet.. Manual MIITS is 19.4 (seems very low @ 12850A). AQA does not register this value. 
	ACR
	
	Problem temporarily solved by increasing the level of helium in current lead).

	Tuesday_11_N
	TBD1
	SSS636 : Warm up quench could not launched. PLC shows fault on QH even though all individual Quench Heaters are charged. 
	MTM
	
	Richard contacted. The cable connecting the quench heaters and the magnet was not present. It has not been connected after Hv test (PT9).

	Wednesday_12_M
	TBA2
	MB2347: Anti-cryostat interlock in cryo application, but no anti-cryostat present on this dipole. Investigation of cryo is going on. 
	ACR
	
	PLC anti-cryostat is not working properly in this cluster

	Wednesday_12_M
	TBC1 
	SSS274: HV insulation (CDBP) Sextupole vs Ground failed. Water condensation has been found on CD EF current leads.
	MTM
	
	OK after having dried the current leads with a fan (by ACR).

	Wednesday_12_A
	TBC2
	SSS141: Corrector powering GH PT5.5, problem. Resistance of the G current lead too high.
	MTM
	
	Fixed after intervention of A. Ballarino dismounting and re-mounting the radiator of the current lead. 

	Thursday_13_M
	TBE2
	MB2353: HW recognition not OK, because of configuration files missing for prep2.2 (IAP @ warm LF).
	MTM
	HWREC
	OK after having restarted TEMA.

	Thursday_13_M
	TBE2
	MB2353 : HV (Prep 3) could not be continued due to problems in SUN server. TBE1 also is affected, as PT2 is due on E1.
	IT
	
	Richard is investigating the issue.

	Thursday_13_M
	TBC1
	SSS274 : TEMA application is hanging while trying to launch training for warming up the magnet, 
	IT
	
	SUN expert has been called.

	Thursday_13_M
	TBB1
	SSS262 : Sun machine is not responding in B bench, while trying to launch HV. 
	IT
	
	SUN Expert informed us to not switch off any application or sun because the problem on E can be extended to other sun. AFS server down and IT upgrade. 

	Thursday_13_N
	TBB1
	SSS262: During launching PT5.3 (corrector powering CD) VS_Fault present on CD and EF power converter.
	AB/PO?
	
	Richard called. Emergency button was pressed.

	Saturday_15_N
	TBD2
	SSS645: Current did not ramp in PT4. Error 222 serial port,

Test was launched again but file is not getting generated. 
Triggers on card 5,7,9,13 existing
	AB/PO
	
	Richard was contacted.
Communication failure with RMD power converter. Richard had to reset the DICO. This can be done after exiting power and server applications.

	Sunday_16_M
	TBA2
	MB2347: Paper traveller not available. ICS was informed. E- Traveller signed
	ICS?
	
	The magnet was removed without paper traveller signed.

	Sunday_16_M
	TBE1
	MB1326: Trigger present on safety matrix.
	ICS
	
	Richard came: One V_tap soldered in the magnet connection by IC, is open at cold. Richard put a bridge.

	Sunday_16_M
	TBC2
	SSS266: problem with TT821. T sensor shorted to the ground.
	MTM
	
	Richard informed. He asked to open a NC.



	Sunday_16_N
	TD2
	SSS645: Fourth training quench:  current ramped up to 2323 A and dropped down to zero without quench & file generation.
	AB/PO?
	
	Slow PA abort. Problem with RMD again?





















